Author's response to reviews

Title: Net benefits: assessing the effectiveness of clinical networks in Australia through qualitative methods

Authors:

Frances C Cunningham (f.cunningham@unsw.edu.au)
Geetha Ranmuthugala (g.ranmuthugala@nsw.edu.au)
Johanna I Westbrook (j.westbrook@unsw.edu.au)
Jeffrey Braithwaite (j.braithwaite@unsw.edu.au)

Version: 4 Date: 29 August 2012

Author's response to reviews: see over
Resubmission of Manuscript Number 7314725997475228 for Implementation Science:
“Net benefits: assessing the effectiveness of clinical networks in Australia through qualitative methods”

We are pleased to resubmit our manuscript titled ‘Net benefits: assessing the effectiveness of clinical networks in Australia through qualitative methods’ as a ‘Research Article’ for publication in Implementation Science.

We thank you for your editorial comments of 28 August 2012 on our revised manuscript and we thank the reviewers for their comments and recommendations for amendments to the manuscript. The revisions suggested by Reviewer 2 and by the editor related to the section on pp 15-16. We have carefully reviewed and made revisions to this section in the Results, in accordance with recommendations of the editor and Reviewer 2. Our changes are shown in track-changes on pp 15-17. We appreciate the opportunity of amending this section of the paper. We believe that this section of the manuscript now flows better, and presents more clearly for the reader.

All authors of this manuscript declare that they have no competing interests. As the corresponding author, I am pleased to resubmit this paper on behalf of all co-authors and confirm that all authors have reviewed and approved the manuscript for submission to Implementation Science.

Yours sincerely

Dr Frances Cunningham
Senior Research Fellow
Centre for Clinical Governance Research
Australian Institute of Health Innovation
University of New South Wales